\chapter{Results}
\label{ch:results}

\section{Comparing images for different $\sigma_1$}

\section{Plots Datafile1.mat}

\subsection{Orginal Plot of Datafile1.mat}
\begin{flushleft}
\begin{figure}[htb]
\begin{center}
\includegraphics[scale=0.25]{Initial Image Datafile1.png} 
\caption{Orginal Image of Datafile1.mat}
\end{center}
\end{figure}
\end{flushleft}


\subsubsection{Images of Datafile1 after each sweep}
\begin{flushleft}
\begin{figure}[htb]
\begin{center}
\includegraphics[scale=0.4]{Pictures of sweeps of datafile1.png} 
\caption{Image at each sweep for $\sigma_1=10$}
\end{center}
\end{figure}
\end{flushleft}


\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of datafile1 (sigma=20).png} 
\caption{Image at each sweep for $\sigma_1=20$}
\end{figure}
\end{flushleft}

\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of datafile1 (sigma=100).png} 
\caption{Image at each sweep for $\sigma_1=100$}
\end{figure}
\end{flushleft}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newpage
\section{Plots Datafile2.mat}

\subsection{Orginal Plot of Datafile2.mat}
\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.25]{Initial Image Datafile2.png} 
\caption{Orginal Image of Datafile2.mat}
\end{figure}
\end{flushleft}


\subsubsection{Images of Datafile2 after each sweep}

\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of DataFile2 (sigma=10).png}  
\caption{Image at each sweep for $\sigma_1=10$}
\end{figure}
\end{flushleft}


\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of Datafile2 (sigma=20).png} 
\caption{Image at each sweep for $\sigma_1=20$}
\end{figure}
\end{flushleft}

\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of Datafile2 (sigma=100).png} 
\caption{Image at each sweep for $\sigma_1=100$}
\end{figure}
\end{flushleft}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newpage
\section{Plots Datafile3.mat}

\subsection{Orginal Plot of Datafile3.mat}
\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.25]{Initial Image Datafile3.png} 
\caption{Orginal Image of Datafile3.mat}
\end{figure}
\end{flushleft}


\subsubsection{Images of Datafile3 after each sweep}
\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of Datafile3 (sigma=10).png}  
\caption{Image at each sweep for $\sigma_1=10$}
\end{figure}
\end{flushleft}


\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of Datafile3 (sigma=20).png} 
\caption{Image at each sweep for $\sigma_1=20$}
\end{figure}
\end{flushleft}

\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of Datafile3 (sigma=100).png} 
\caption{Image at each sweep for $\sigma_1=100$}
\end{figure}
\end{flushleft}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newpage
\section{Plots Datafile4.mat}

\subsection{Orginal Plot of Datafile4.mat}
\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.25]{Initial Image Datafile4.png} 
\caption{Orginal Image of Datafile4.mat}
\end{figure}
\end{flushleft}


\subsubsection{Images of Datafile4 after each sweep}
\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of Datafile4 (sigma=10).png}  
\caption{Image at each sweep for $\sigma_1=10$}
\end{figure}
\end{flushleft}


\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of Datafile4 (sigma=20).png} 
\caption{Image at each sweep for $\sigma_1=20$}
\end{figure}
\end{flushleft}

\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of Datafile4 (sigma=100).png} 
\caption{Image at each sweep for $\sigma_1=100$}
\end{figure}
\end{flushleft}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newpage
\section{Plots Datafile5.mat}

\subsection{Orginal Plot of Datafile5.mat}
\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.25]{Initial Image Datafile5.png} 
\caption{Orginal Image of Datafile5.mat}
\end{figure}
\end{flushleft}


\subsubsection{Images of Datafile5 after each sweep}
\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of Datafile5 (sigma=10).png}  
\caption{Image at each sweep for $\sigma_1=10$}
\end{figure}
\end{flushleft}


\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of Datafile5 (sigma=20).png} 
\caption{Image at each sweep for $\sigma_1=20$}
\end{figure}
\end{flushleft}

\begin{flushleft}
\begin{figure}[htb]
\includegraphics[scale=0.4]{Pictures of sweep5 of Datafile5 (sigma=100).png} 
\caption{Image at each sweep for $\sigma_1=100$}
\end{figure}
\end{flushleft}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%